Author’s response to reviews

Title: Comparison of brain connectomes by MRI and genomics and its implication in Alzheimer’s disease

Authors:

Young Jae Woo (youngjae.woo@mssm.edu)
Panos Roussos (panagiotis.roussos@mssm.edu)
Vahram Haroutunian (vahram.haroutunian@mssm.edu)
Pavel Leon Katsel (pavel.katsel@mssm.edu)
Samuel Gandy (samuel.gandy@mssm.edu)
Eric E Schadt (eric.schadt@mssm.edu)
Jun Zhu (jun.zhu@mssm.edu;junzhu_99@yahoo.com)

Version: 2 Date: 22 Oct 2019

Author’s response to reviews:

All responses are detailed in the "response to reviewer comments" letter.